Author’s response to reviews

Title: Cognitive Behaviour Therapy in Medication-Treated Adults with ADHD and Persistent Symptoms: A randomized controlled trial.

Authors:

Brynjar Emilsson (brynjare@landspitali.is)
Gisli Gudjonsson (gisli.gudjonsson@kcl.ac.uk)
Jon F Sigurdsson (jonfsig@landspitali.is)
Gisli Baldursson (gislib@landspitali.is)
Emil Einarsson (emile@landspitali.is)
Halldora Olafsdottir (haldola@landspitali.is)
Susan Young (susan.young@kcl.ac.uk)

Version: 3 Date: 18 July 2011

Author’s response to reviews: see over
Dr Sabina Alam,
Senior Scientific Editor,
BMC-series journals,

Dear Dr Alam,

Ref. MS:1540603760528406
Cognitive Behaviour Therapy in Medication-Treated Adults with ADHD and Persistent Symptoms: A randomized controlled trial.

Following acceptance in principle of the above manuscript for publication in BMC Psychiatry, please find attached the latest version. The paper now includes the following sentence at the beginning of the Procedure section of the Method (page 11), as requested:

The study was conducted in line with international guidelines, following ethical approval by the Icelandic Bioethics Committee on 01/09/2008, reference number 08-095-S1.

We look forward to seeing this article in press.

Yours sincerely,

Dr Susan Young.